**Author's response to reviews**

**Title:** Do existing mechanisms contribute to improvements in care coordination across levels of care in health services networks? Opinions of the health personnel in Colombia and Brazil

**Authors:**

Ingrid Vargas (ivargas@consorci.org)  
Amparo Susana Mogollón-Pérez (amparo.mogollon@urosario.edu.co)  
Pierre De Paepe (pdpaeppe@itg.be)  
Maria Rejane Ferreira da Silva (rejane@cpqam.fiocruz.br)  
Jean-Pierre Unger (JPUnger@itg.be)  
María Luisa Vázquez (mlvazquez@consorci.org)

**Version:** 4 **Date:** 28 April 2015

**Author's response to reviews:** see over
Title: Do existing mechanisms contribute to improvements in care coordination across levels of care in health services networks? Opinions of health personnel in Colombia and Brazil

Dear Mr. Auro,

We are pleased to send you the revised version of our paper “Do existing mechanisms contribute to improvements in care coordination across levels of care in health services networks? Opinions of health personnel in Colombia and Brazil”, which contains the editorial revisions required:

- Add the full names and affiliations of all the ethics committees that approved the study to the ethics statement.
  We have added this information in the ethical considerations section.

- State in the manuscript whether the informed consent was written or verbal.
  Free and written informed consent was obtained from every interviewee. We have added this information in the ethical considerations section.

- Delete the funding section and add funding info to Acknowledgement.
  We have moved the funding information into the manuscript as it is requested.

Once carried out the editorial revisions, we hope that our manuscript will now be considered suitable for publication in your journal.

Kind regards,

Ingrid Vargas
Amparo S. Mogollón
Rejane Ferreira-da-Silva
Pierre de Paepe
Jean-Pierre Unger
M- Luisa Vázquez